
-----------------------------------------------------------------------------
          Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
       Wed Oct 11 16:48:35 GMT 2017
Year: 17  Doy: 284
Observer: berkey
WEATHER COMMENT: berkey: Wed Oct 11 16:49:45 GMT 2017
Temp: 49.5f, Humidity: 10%, Pressure: 28.583in, Wind: 5mph from 150degs, Skies: clear
___end___
GENERAL COMMENT BY Observer: Wed Oct 11 21:02:18 GMT 2017
Test
___end___
1 17:22:41 GMT 2017 SGS Alignment complete
Wed Oct 11 17:23:05 GMT 2017 Kcor Focus/alignment program exited
Wed Oct 11 17:22:53 GMT 2017 CoMP occulter has been re-centered
Wed Oct 11 17:22:54 GMT 2017 COMP Start Patrol on cookbook: synoptic-00010.cbk
Wed Oct 11 17:27:48 GMT 2017 KCOR Start Synoptic Patrol
Wed Oct 11 20:04:26 GMT 2017 COMP End Patrol
Wed Oct 11 20:04:27 GMT 2017 COMP Start Patrol on cookbook: waves-00001.cbk
Wed Oct 11 21:00:33 GMT 2017 KCOR End Patrol
Wed Oct 11 21:02:13 GMT 2017 SGS Alignment complete
Wed Oct 11 21:03:22 GMT 2017 Kcor Focus/alignment program exited
Wed Oct 11 21:08:36 GMT 2017 KCOR Start Synoptic Patrol
Wed Oct 11 21:08:37 GMT 2017 KCOR End Patrol
Wed Oct 11 21:08:38 GMT 2017 KCOR Start Calibration script: c:\kcor\mlso-calibration22deg-20150323.ini
Wed Oct 11 21:23:58 GMT 2017 KCOR End Calibration Script
Wed Oct 11 21:24:15 GMT 2017 KCOR Start Synoptic Patrol
Wed Oct 11 21:24:16 GMT 2017 KCOR Start Synoptic Patrol
Wed Oct 11 21:36:44 GMT 2017 COMP End Patrol
Wed Oct 11 21:36:45 GMT 2017 COMP Start Patrol on cookbook: synoptic-00010.cbk
Wed Oct 11 22:19:29 GMT 2017 CoMP Paused for clouds
Wed Oct 11 22:21:42 GMT 2017 CoMP Restarted from pause
Wed Oct 11 22:23:32 GMT 2017 KCOR End Patrol
Wed Oct 11 22:23:22 GMT 2017 CoMP Paused for clouds
Wed Oct 11 22:41:53 GMT 2017 CoMP Restarted from pause
Wed Oct 11 22:43:34 GMT 2017 SGS Alignment complete
Wed Oct 11 22:44:08 GMT 2017 Kcor Focus/alignment program exited
Wed Oct 11 22:44:21 GMT 2017 CoMP Paused for clouds
Wed Oct 11 22:54:20 GMT 2017 CoMP Restarted from pause
Wed Oct 11 23:00:57 GMT 2017 KCOR Start Synoptic Patrol
Wed Oct 11 23:19:01 GMT 2017 KCOR End Patrol
Wed Oct 11 23:19:37 GMT 2017 CoMP Paused for clouds
**GENERAL PROBLEM COMMENT BY berkey** : Thu Oct 12 01:18:38 GMT 2017
For some reason the logger computer is having a problem with its connection to the data computer.  Not sure why but on the
 first read of the log the computer cached the old version of log so it never saw the changes made on the server.   Added 
a kudge to the logging software to disconnect/reconnect to the server on every log read which seems to have fixed the issu
e with reading log file updates; but the GUI feels a lot slower with the kludge appled.
___end___



Thu Oct 12 02:15:01 GMT 2017 CoMP Restarted from pause
Thu Oct 12 02:15:01 GMT 2017 COMP End Patrol
GENERAL COMMENT BY berkey: Thu Oct 12 02:15:42 GMT 2017
Nice morning but afternoon skies got dominated by aersol and cirus.
___end___


